Author's response to reviews

Title: Corticosteroid therapy in regressive autism: A retrospective study of effects on the Frequency Modulated Auditory Evoked Response (FMAER), language, and behavior

Authors:

Frank H Duffy MD (fhd@sover.net)
Aditi Shankardass PhD (aditi.shankardass@childrens.harvard.edu)
Gloria B Mcanulty PhD (gloria.mcanulty@childrens.harvard.edi)
Yaman Z Eksioglu MD (eksiogly@upstate.edu)
David Coulter MD (david.coulter@childrens.harvard.edi)
Alexander Rotenberg MD (alexander.rotenberg@childrens.harvard.edu)
Heidelise Als PhD (heidelise.als@childrens.harvard.edu)

Version: 7
Date: 2 April 2014

Author's response to reviews:

see BMC Medicine